Author's response to reviews

Title: Role of LPAR3, PKC and EGFR in LPA-induced cell migration in oral squamous carcinoma cells

Authors:

Ingvild J Brusevold (i.j.brusevold@medisin.uio.no)
Ingun H Tveteraas (i.h.tveteraas@medisin.uio.no)
Monica Aasrum (monica.aasrum@medisin.uio.no)
John Ødegård (john.odegard@medisin.uio.no)
Dagny L Sandnes (d.l.sandnes@medisin.uio.no)
Thoralf Christoffersen (thoralf.christoffersen@medisin.uio.no)

Version: 2
Date: 6 December 2013

Author's response to reviews:

Cover letter
December 6th 2013
Please find enclosed our manuscript which we resubmit today with some minor changes after request of the Journal Editorial Office.
To the manuscript, we have added the following sections:
• Competing interests
• Authors’ contributions
• Acknowledgements
In addition, we have made minor formal changes in the figure legends.

Sincerely,
Ingvild J. Brusevold